Author’s response to reviews

Title: Open ventral hernia repair with a composite ventral patch - final results of a multicenter prospective study

Authors:

Frederik Berrevoet (frederik.berrevoet@ugent.be)

Carl Doerhoff (LRoberts@crmc.org)

Filip Muysoms (Flip.muysoms@gmail.com)

Steven Hopson (sbhopson1@gmail.com)

Gallinella Muzi (marcog.muzi@gmail.com)

S. Nienhuijs (simon.nienhuijs@cze.nl)

Eric Kullman (Eric.Kullman@lio.se)

Tim Tollens (tollenstim@hotmail.com)

Mark Schwartz (markschwartz22@gmail.com)

Karl Leblanc (karl.leblanc@ololrmc.com)

Vic Velanovich (vvelanov@health.usf.edu)

Lars Nannestad Jørgensen (larsnjorgensen@hotmail.com)

Version: 1 Date: 02 Apr 2019

Author’s response to reviews:

In response to the reviewer comments of our manuscript the authors have made the following changes (responses outlined in <<>>). If there is anything further I can assist with please let me know.

Best,

Nicholas Paquette on behalf of Dr. Berrevoet

Editor's comments:
1. We note that the current submission contains some textual overlap with other previously published works, in particular:


This overlap exists throughout your manuscript.

While we understand that this is work that you have previously published, and some of the same ideas are contained in these publications, please be aware that we cannot condone the use of text from previously published work.

Please be informed that we cannot proceed with handling your manuscript before this issue is resolved, and the sections of text in question have been reformulated.

<<The overlapping text has been reviewed and revised>>

2. We note that the email addresses provided for Carl Doerhoff, S. Nienhuijs and Steven Hopson in the manuscript do not match those provided in our editorial submissions system Please amend one so that both match.

<<These email addresses have been updated in the system.>>

3. Please remove the funding information from the competing interests section, please include this in the funding section instead. If you do not have any competing interests, please state "not applicable" instead.
4. Please remove the funding information from the acknowledgements section. If you do not have any acknowledgements, please state "not applicable instead".

<<This information has been removed from the acknowledgements section. >>